Author's response to reviews

Title: Estimates of cancer incidence, mortality and survival in Aboriginal people from NSW, Australia

Authors:

Stephen Morrell (stephen.morrell@cancerinstitute.org.au)
Hui You (hui.you@cancerinstitute.org.au)
Deborah Baker (deborah.baker@cancerinstitute.org.au)

Version: 2 Date: 21 September 2011

Author's response to reviews:

I inadvertently left a legend off figure 4. Subsequently I need to upload an updated figure 4. Consequently, I need also to upload a slightly revised version of the main manuscript to reflect a change in the position of the 'Incidence' and 'Mortality' labels of this figure since the figure is a slightly different size.

thanks